Author's response to reviews

Title: Do personalised e-mail invitations improve recruitment of breast cancer survivors into a web-based behaviour change intervention?

Authors:

Camille E Short (camille.short@adelaide.edu.au)
Amanda L Rebar (a.rebar@cqu.edu.au)
Corneel Vandelanotte (c.vandelanotte@cqu.edu.au)

Version: 3 Date: 2 March 2015

Author's response to reviews:

Dear Arlene,

Thank you for inviting us to resubmit our article as a correspondence article. We have reformatted our article based on the information provided. The only section requiring major changes was the abstract - which we have reduced to 100 words.

Thank you for considering our manuscript.

Kind regards,

Camille Short
NHMRC ECR Fellow
University of Adelaide, Australia